**Author's response to reviews**

**Title:** Demographic, psychosocial, and genetic risk associated with smokeless tobacco use among Mexican heritage youth

**Authors:**

Anna V Wilkinson (anna.v.wilkinson@uth.tmc.edu)
Laura M Koehly (koehlyl@mail.nih.gov)
Elizabeth A Vandewater (elizabeth.a.vandewater@uth.tmc.edu)
Robert K Yu (rkyu@mdanderson.org)
Susan P Fisher-Hoch (Susan.P.Fisher-Hoch@uth.tmc.edu)
Alexander V Prokhorov (apokhor@mdanderson.org)
Harold W Kohl (Harold.W.Kohl@uth.tmc.edu)
Margaret R Spitz (spitz@bcm.edu)
Sanjay Sanjay Shete (sshete@mdanderson.org)

**Version:** 2  
**Date:** 19 November 2014

**Author's response to reviews:** see over
November 19, 2014

Dear BMC Editorial Team:

Thank you for your email. I am sorry I forgot all of the formatting requirements in my eagerness to submit!

I have made the following changes:

- **Title Page:**
  On the title page, I added the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.

- **Line Numbering:**
  Added

- **Competing Interests:**
  Added

- **Authors’ Contributions:**
  Added

- I moved the Acknowledgement section before the references section.

Thank you for your email and sorry for the inconvenience.

Best, Anna